**Author's response to reviews**

**Title:** Selection Criteria for Radiofrequency Ablation for Colorectal Liver Metastases in the Era of Effective Systemic Therapy: a Clinical Score based Proposal

**Authors:**

Axel Stang (a.stang@asklepios.com)
Karl J Oldhafer (k.oldhafer@asklepios.com)
Hauke Weilert (h.weilert@asklepios.com)
Handan Keles (handankeles@yahoo.de)
Marcello Donati (mdonati@unict.it)

**Version:** 6 **Date:** 16 June 2014

**Author's response to reviews:** see over
Selection criteria for radiofrequency ablation for colorectal liver metastases in the era of effective systemic therapy: a clinical score based proposal

Dear BMC Cancer Editorial Team,

We would like to submit our revised manuscript (corrected, grammatical errors eliminated). Following your recommendation, we asked a native English speaking colleague to copyedit our paper. In fact, our native English speaking colleague found several grammatical errors in our last version, which are eliminated in our revised manuscript.

Payment of the article processing charge (EUR 1.595 plus administration charge) will be executed immediately after receipt of invoice. There are 2 alternatives (to send the invoice):
1. Postal address: Asklepios ProResearch / Lohmühlenstraße 5 / Haus J / 20099 Hamburg / Germany / Konto Dr. Stang or
2. Email address (much easier): i.karasu@asklepios.com (invoice payment will follow immediately)

If you have any problems with our corrected final version, please do not hesitate to contact us.

With best wishes,

Dr. Axel Stang, MD (corresponding author, on behalf of the co-authors)

Department of Hematology, Oncology, and Palliative Care
Aklepios Hospital Barmbek
Rübenkamp 220, 22291 Hamburg (Germany)
Phone: +49 (0) 40 18 18 82 – 3830
FAX: +49 (0) 40 18 18 82 – 3309
E-mail: a.stang@asklepios.com